#### Washington University School of Medicine Digital Commons@Becker

2020-Current year OA Pubs

**Open Access Publications** 

11-1-2023

## Dominant negative variants in KIF5B cause osteogenesis imperfecta via down regulation of mTOR signaling

Ronit Marom Anika Lindsey Jacob Lesinski Michael L. Nonet Jian Chen

See next page for additional authors

Follow this and additional works at: https://digitalcommons.wustl.edu/oa\_4

Part of the Medicine and Health Sciences Commons Please let us know how this document benefits you.

#### Authors

Ronit Marom, Anika Lindsey, Jacob Lesinski, Michael L. Nonet, Jian Chen, Dustin Baldridge, Gary A. Silverman, Stephen C. Pak, and et al.



#### GOPEN ACCESS

**Citation:** Marom R, Zhang B, Washington ME, Song I-W, Burrage LC, Rossi VC, et al. (2023) Dominant negative variants in *KIF5B* cause osteogenesis imperfecta via down regulation of mTOR signaling. PLoS Genet 19(11): e1011005. https://doi.org/10.1371/journal.pgen.1011005

Editor: Gregory S. Barsh, HudsonAlpha Institute for Biotechnology, UNITED STATES

Received: May 29, 2023

Accepted: October 3, 2023

Published: November 7, 2023

**Copyright:** © 2023 Marom et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data generated or analyzed during this study are included in the manuscript and supporting files. Source data file has been provided for Fig 6A that includes a full list of differentially expressed genes resulting from RNA-seq analysis of human fibroblasts. Novel variants reported in this manuscript were deposited to ClinVar (National Center for Biotechnology Information. ClinVar; [VCV002580234.1], https:// www.ncbi.nlm.nih.gov/clinvar/variation/ VCV002580234.1). Research sequencing data was deposited to dbGaP (phs001232.v5.p2). https:// RESEARCH ARTICLE

# Dominant negative variants in *KIF5B* cause osteogenesis imperfecta via down regulation of mTOR signaling

Ronit Marom<sup>1,2</sup>, Bo Zhang<sup>3</sup>, Megan E. Washington<sup>1</sup>, I-Wen Song<sup>1</sup>, Lindsay C. Burrage<sup>1,2</sup>, Vittoria C. Rossi<sup>1,2</sup>, Ava S. Berrier<sup>1</sup>, Anika Lindsey<sup>3</sup>, Jacob Lesinski<sup>3</sup>, Michael L. Nonet<sup>4</sup>, Jian Chen<sup>5</sup>, Dustin Baldridge<sup>3</sup>, Gary A. Silverman<sup>3</sup>, V. Reid Sutton<sup>1,2</sup>, Jill A. Rosenfeld<sup>1</sup>, Alyssa A. Tran<sup>1</sup>, M. John Hicks<sup>2,6</sup>, David R. Murdock<sup>1</sup>, Hongzheng Dai<sup>1</sup>, MaryAnn Weis<sup>7</sup>, Shalini N. Jhangiani<sup>1</sup>, Donna M. Muzny<sup>1</sup>, Richard A. Gibbs<sup>1</sup>, Richard Caswell<sup>6</sup>, Carrie Pottinger<sup>9</sup>, Deirdre Cilliers<sup>10</sup>, Karen Stals<sup>8</sup>, Undiagnosed Diseases Network<sup>1</sup>, David Eyre<sup>7</sup>, Deborah Krakow<sup>11</sup>, Tim Schedl<sup>5‡</sup>, Stephen C. Pak<sup>3‡</sup>, Brendan H. Lee<sup>1,2‡</sup>\*

Department of Molecular and Human Genetics, Baylor College of Medicine, Houston, Texas, United States of America, 2 Texas Children's Hospital, Houston, Texas, United States of America, 3 Department of Pediatrics, Washington University School of Medicine, St Louis, Missouri, United States of America,
Department of Neuroscience, Washington University School of Medicine, St Louis, Missouri, United States of America, 5 Department of Genetics, Washington University School of Medicine, St Louis, Missouri, United States of America, 6 Department of Pathology & Immunology, Baylor College of Medicine, Houston, Texas, United States of America, 7 Department of Orthopaedics and Sports Medicine, University of Washington, Seattle, Washington, United States of America, 8 Exeter Genomics Laboratory, Royal Devon University Healthcare NHS Foundation Trust, Exeter, United Kingdom, 9 All Wales Medical Genomics Service, Wrexham Maelor Hospital, Wrexham, UK, 10 Oxford Centre for Genomic Medicine, Oxford University Hospitals NHS Foundation Trust, Oxford, United Kingdom, 11 Human Genetics, Obstetrics & Gynecology, Orthopedic Surgery, University of California, Los Angeles, California, United States of America

- These authors contributed equally to this work.
- ‡ These authors are joint senior authors on this work.
- <sup>¶</sup> Membership of Undiagnosed Diseases Network is provided in Supporting Information file S1 Acknowledgement.
- \* blee@bcm.edu

#### Abstract

#### Background

Kinesin motor proteins transport intracellular cargo, including mRNA, proteins, and organelles. Pathogenic variants in kinesin-related genes have been implicated in neurodevelopmental disorders and skeletal dysplasias. We identified *de novo*, heterozygous variants in *KIF5B*, encoding a kinesin-1 subunit, in four individuals with osteogenesis imperfecta. The variants cluster within the highly conserved kinesin motor domain and are predicted to interfere with nucleotide binding, although the mechanistic consequences on cell signaling and function are unknown.

#### Methods

To understand the *in vivo* genetic mechanism of *KIF5B* variants, we modeled the p.Thr87lle variant that was found in two patients in the *C. elegans* ortholog, *unc-116*, at the corresponding position (Thr90lle) by CRISPR/Cas9 editing and performed functional analysis. Next, we

www.ncbi.nlm.nih.gov/projects/gap/cgi-bin/study. cgi?study\_id=phs001232.v5.p2. Numerical data that underlies graphs or summary statistics is provided as supporting information.

Funding: RM was supported by the Michael Geisman Fellowship of the Osteogenesis imperfecta foundation, the Lawrence Family Bone Disease Program of Texas, the BCM Chao Physician- Scientist award of the Ting Tsung and Wei Fong Chao Foundation, and through the NIH/ NIGMS T32GM07526. Research reported in this manuscript was also supported by the NIH Common Fund, through the Office of Strategic Coordination/Office of the NIH Director under award number U01 HG007709 (BL), U54 NS108251 (TS), U01 HG007703 and U01 HG007942. Research reported in this publication was supported by the Eunice Kennedy Shriver National Institute of Child Health & Human Development of the National Institutes of Health under award number P50HD103555 for the Baylor College of Medicine Intellectual and Developmental Disabilities Research Center (IDDRC). Funding was also provided by the Children's Discovery Institute, St Louis Children's Hospital Foundation (GAS and SCP) and R01 GM100756 (TS). The content of this manuscript is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

studied the cellular and molecular consequences of the recurrent p.Thr87lle variant by microscopy, RNA and protein analysis in NIH3T3 cells, primary human fibroblasts and bone biopsy.

#### Results

*C. elegans* heterozygous for the *unc-116* Thr90lle variant displayed abnormal body length and motility phenotypes that were suppressed by additional copies of the wild type allele, consistent with a dominant negative mechanism. Time-lapse imaging of GFP-tagged mito-chondria showed defective mitochondria transport in *unc-116* Thr90lle neurons providing strong evidence for disrupted kinesin motor function. Microscopy studies in human cells showed dilated endoplasmic reticulum, multiple intracellular vacuoles, and abnormal distribution of the Golgi complex, supporting an intracellular trafficking defect. RNA sequencing, proteomic analysis, and bone immunohistochemistry demonstrated down regulation of the mTOR signaling pathway that was partially rescued with leucine supplementation in patient cells.

#### Conclusion

We report dominant negative variants in the *KIF5B* kinesin motor domain in individuals with osteogenesis imperfecta. This study expands the spectrum of kinesin-related disorders and identifies dysregulated signaling targets for *KIF5B* in skeletal development.

#### Author summary

Kinesin-related gene defects are associated with multi-systemic disorders, also known as "Kinesinopathies". Heterozygous variants in *KIF5B*, the gene encoding the Kinesin-1 heavy chain, have been implicated in neurodevelopmental delay, skeletal dysplasia, myop-athy, and dysmorphic features. We identified *de novo*, heterozygous missense variants in *KIF5B* kinesin motor domain in four individuals with osteogenesis imperfecta. Findings in animal model and cell studies point to a dominant negative mechanism, an intracellular trafficking defect, and alteration of mTOR signaling. The study identifies *KIF5B* variants as a novel cause for osteogenesis imperfecta and suggests that mTOR signaling may be potentially targeted for future therapy in *KIF5B*-related disorder.

#### Introduction

Kinesin superfamily of motor proteins (KIFs) play an essential role in fundamental cellular processes, including endocytosis, cell cycle, autophagy and ciliogenesis [1–3]. Within this family, kinesin-1 utilizes ATP to transport cargo, such as mRNA, secretory vesicles, mitochondria, and lysosomes along oriented microtubule tracks [4–7]. Kinesin-1 is a tetramer composed of two microtubule-associated heavy chains and two cargo-binding light chain molecules [8,9]. The kinesin-1 heavy chain is encoded by either of three *KIF5* genes: *KIF5B*, which is ubiquitously expressed, or its neuron-specific heavy chain isoforms *KIF5A* and *KIF5C* [3,8]. Homozygous knockout of *Kif5b* in mice (*Kif5b*<sup>-/-</sup>) results in severe growth retardation and embryonic lethality at E. 9.5–11.5 [10]. Conditional deletion of the murine *Kif5b* in pancreatic beta cells in *Kif5b*<sup>fl/-</sup>;RIP2-Cre mice is associated with a defect in insulin secretion, decreased

islet size, and diabetes [11]. *Kif5b*<sup>fl/-</sup>; Pax2-Cre mice that have a genetic deletion of *Kif5b* in skeletal muscle exhibit impaired myofibril assembly and muscular dystrophy [12]. Genetic deletion of *Kif5b* in the chondrocyte lineage via Col2a1-Cre was associated with a chondrocyte differentiation defect and disruption of the growth plate ultrastructure, leading to growth retardation [13]. Consistent with these findings, *kif5b* loss of function in zebrafish resulted in abnormal craniofacial development, with deficits in muscle development, chondrogenesis and skeletal mineralization [14]. While these studies highlight the critical role of *KIF5B* in development across tissues and species, downstream signaling and cellular consequences are poorly understood.

Pathogenic variants in KIF-related genes, also known as "kinesinopathies" [15], have been associated with variable human phenotypes, including neurodevelopmental disorders and skeletal dysplasias [16–22]. Itay *et al.* reported heterozygous missense variants in *KIF5B* in individuals that were clinically diagnosed with kyphomelic dysplasia and presented with short stature, osteoporosis, fractures, bone deformities, and variable developmental delay [23]. Flex *et al.* reported heterozygous pathogenic variants in *KIF5B* in individuals with developmental delay, skeletal myopathy, and adult-onset cardiomyopathy but no features of skeletal dysplasia [24].

Here, we identified heterozygous, *de novo* variants in *KIF5B* in four unrelated individuals with osteogenesis imperfecta, including a recurrent variant p.Thr87Ile that resides in the highly conserved nucleotide-binding P-loop domain. The recurrent variant p.Thr87Ile was modeled *in vivo* in *C. elegans* and studied *in vitro* to assess the cellular and molecular alterations downstream of *KIF5B* variants. This study expands the clinical spectrum of *KIF5B*-related disorder and informs on the mechanism of pathogenic variants in kinesin-1 in skeletal dysplasia.

#### Results

### *De novo*, heterozygous variants in *KIF5B* were identified in individuals with osteogenesis imperfecta

We performed a clinical and molecular study in four individuals with an overlapping phenotype characterized by hypomineralization of bones, multiple fractures, and skeletal deformities (Fig 1A–1C). Exome sequencing identified *de novo*, heterozygous variants in *KIF5B* that reside within the kinesin motor domain (Fig 1D and Table 1). All variants identified substitute highly conserved amino acid residues and are predicted to destabilize the nucleotide binding pocket that is critical for the motor activity of the protein (Fig 1E).

Proband 1 is a 20-year-old female with a clinical diagnosis of osteogenesis imperfecta of moderate severity. She has a history of recurrent fractures with minor or no trauma, low bone mineral density, short stature, scoliosis and dentinogenesis imperfecta (Fig 1A–1C). In addition to the skeletal features, she has a history of bilateral mixed hearing loss, ptosis, an atrial septal defect, and a splenic cyst. She had mild developmental delay which resolved, and her academic performance in school was above average. Clinical genetic testing, including chromosome microarray analysis and OI gene panel, was non-diagnostic. Collagen analysis in skin fibroblasts showed normal gel migration and prolyl hydroxylation and ruled out a qualitative or quantitative collagen defect (S1 Fig). Research trio exome sequencing identified a heterozygous, *de novo* variant, c.260C>T, p.Thr87Ile (NM\_004521.2) in *KIF5B* that was confirmed by Sanger sequencing. RNA sequencing in fibroblasts demonstrated that both the wild type and mutant alleles are expressed in the proband (sequencing detected the variant in 21/57, or 37% of the reads in proband's fibroblasts). Proband 2 is a 30-year-old female with a clinical diagnosis of osteogenesis imperfecta. She has a history of low bone mineral density, multiple fractures, scoliosis and dentinogenesis imperfecta. She is also noted to have bilateral mixed



**Fig 1. Recurrent heterozygous variants in** *KIF5B* **cause osteogenesis imperfecta. (A-C) Skeletal radiographs of Proband 1. (A) Osteopenia with gracile ribs and fracture of clavicle (arrow) at age 1 week; (B) Healing fractures and bowing of long bones, early childhood; (C) Scoliosis, adolescent years. (D) Graphic illustration of** *KIF5B* **gene, highlighting pathogenic variants reported here (red, bold) and elsewhere [23,24,83], within functional domains. In red: Pathogenic variants associated with skeletal dysplasia (within the kinesin motor domain, the P-loop and the switch loops create and stabilize the nucleotide binding pocket). The amino acids sequence within the P loop domain, outlined in the figure, is highly conserved between species. In black: Pathogenic variants outside the P loop and switch loop domains are not associated with a skeletal phenotype. (E) KIF5B nucleotide binding and potential mechanism of variants reported here: In the ADP-bound form of KIF5B (left), the backbone of Gly90 (bright green) makes hydrogen bonds (green broken lines) with ADP, while the backbone of Thr87 (yellow) makes hydrogen bonds (yellow broken lines) with Glu236. Note that, in this form, Thr195 (cyan) is distant from the bound nucleotide with the sidechain directed outward of the protein surface. In the ATP-bound form of KIF5B (right), Gly90 bonding to ATP is retained while the movement of the switch I loop brings Thr195 and Asn198 (blue) sidechains into proximity with the** 

Е

nucleotide, allowing multiple potential hydrogen bonds with phosphate groups (green broken lines) and stabilization of the conformational switch.

https://doi.org/10.1371/journal.pgen.1011005.g001

hearing loss and ptosis. She had a typical development with no history of cognitive impairment. Research whole genome sequencing detected a heterozygous variant, c.260C>T, p.Thr87Ile (NM\_004521.2) in *KIF5B*, same as found in proband 1. Probands 3 and 4 were both products of conception and pregnancies were terminated due to multiple congenital anomalies. The clinical findings, including hypomineralized bones, *in utero* fractures and bone deformities, were suspicious of osteogenesis imperfecta. Exome sequencing identified *de novo*, heterozygous variants: c.269G>C, p.Gly90Ala (proband 3), and c.584C>A, p.Thr195Lys (proband 4) in *KIF5B* (NM\_004521.2). None of the *KIF5B* variants found in proband 1–4 were reported in gnomAD, a control population database [25]. The clinical and molecular findings in probands 1–4 are summarized in Table 1.

## The Thr90Ile variant is damaging to UNC-116 function, acting as a dominant negative allele

To assess the functional impact of the p.Thr87Ile variant on KIF5B activity, we modeled the variant in the C. elegans ortholog, unc-116. The UNC-116 motor domain shares 71% amino acid identity (84% similarity) with KIF5B. Moreover, the proband variant resides in the P-loop of the ATP binding domain, which is 100% conserved in UNC-116 (S2 Fig). This allowed us to use CRISPR/Cas9 genome editing to precisely knock-in the Thr87Ile variant into the corresponding Thr90 residue in unc-116. As part of our editing strategy, we introduced synonymous changes to block Cas9 re-cleavage and generated a restriction enzyme site for allele genotyping. Therefore, as a control, we also generated the unc-116(T90T) edit that only contains the synonymous and restriction enzyme changes but not the variant edit. In addition, we obtained a deletion allele, *unc-116(del)*, derived from *unc-116(gk5722)* [26] for comparison. Animals homozygous for the unc-116(T90T) control developed normally and were indistinguishable from the wild-type VC2010 parental line (Fig 2A and 2B). In contrast, the unc-116 (T90I) variant homozygotes arrested (and died) at the first larval (L1) stage (Fig 2A and 2B). Similarly, most of the *unc-116(del*) homozygotes arrested at the L1 stage, although approximately 10% of these animals eventually developed into adults (escaping adults) by 7 days after egg-lay (Fig 2A and 2B). These data indicated that the T90I variant is damaging to UNC-116 function and that the variant animals had a more severe defect than unc-116(del) null mutants.

Unlike the variant homozygotes, variant heterozygotes are viable and grow to be fertile adults. To determine whether the variant heterozygotes display dominant phenotypes, we quantified the animal's body length and swimming speed using the WormLab 2019.1.1 (MBF Bioscience LLC). The body length of *unc-116(del)* heterozygotes was normal and not significantly different to that of *unc-116(T90T)* control heterozygotes (Fig 2C). However, the *unc-116(T90I)* variant heterozygotes had body lengths that were significantly shorter (Fig 2C). In terms of swimming, the *unc-116(T90T)* control and *unc-116(del)* heterozygotes thrashed similarly to the wild-type controls (Fig 2D). However, the thrashing speed of the *unc-116(T90I)* variant heterozygotes was markedly slower than the controls (Fig 2D) and S1 and S2 Movies). These results indicated that the *unc-116(T90I)* heterozygotes displayed dominant phenotypes not observed in deletion heterozygotes, suggesting that the T90I is a dominant negative or a hyperactive variant.

Next, we employed classic gene dosage studies [27,28] to better understand the nature of the *unc-116(T90I)* variant; if the variant gives rise to a dominant negative gene product, then

	Proband 1	Proband 2	Proband 3	Proband 4
<i>KIF5B</i> variant (NM_004521.2)	c.260C>T p.Thr87Ile	c.260C>T p.Thr87Ile	c.269G>C p.Gly90Ala	c.584C>A p.Thr195Lys
Inheritance	De novo	De novo	De novo	De novo
CADD score	28.7	28.7	28.7	28.9
Sex	Female	Female	Female	Male
Age at most recent evaluation (years)	20	30	23+2/40 weeks gestation (POC)	20/40 weeks gestation (POC)
Pregnancy & Birth				
IUGR/SGA	Yes (BW 1673 grams)	Yes (BW 2353 grams)	Yes (BW 480 grams, 5% ile)	Yes (BW 265.3 grams, 7%ile)
Gestational age at birth	40 weeks	38 weeks	Pregnancy terminated at 23+2/40 weeks	Pregnancy terminated at 20/40 weeks
Findings on prenatal ultrasound	unknown	oligohydramnios	Short long bones with bowed femurs, <i>in utero</i> fractures	Abnormal ribs, short long bones, small chest size, hypomineralisation, tetralogy of Fallot, absent pulmonary valve, absent ductus venosus, micrognathia
Growth				
Growth parameters on most recent evaluations (Z score or normal range)	Height 134 cm (Z = -4.78) weight 28.2 kg (Z = -7.6)	Height 91.4 cm (Z = -10.8) weight 27.2 kg (Z = -9.2)	Weight 480 g (510 +/- 97 g) Crown-heel 25.5cm (29.1 +/- 2cm)	Weight 265.3g (307 +/- 47g) Crown-heel 20.5cm (24.4 +/- 1.8cm)
Microcephaly	Yes (FOC Z = -3.12)	No	No	No
OI Features				
Osteopenia	Yes	Yes	Yes	Yes
Multiple fractures	Yes	Yes	<i>In utero</i> bilateral fractures of femur, radius and ulna, and multiple rib fractures	No
Bone deformities	Yes	Yes	Yes (Angulation of humeri, right tibia and left fibula)	Yes (bent and short long bones, hypoplastic and unossified vertebrae and pelvic bones)
Scoliosis	Yes	Yes	No	No
Dentinogenesis imperfecta	Yes +Hypodontia, malocclusion	Yes	N/A	N/A
Hearing loss	Bilateral mild to moderate mixed hearing loss	Bilateral moderate to severe mixed hearing loss	N/A	N/A
Pulmonary symptoms	Restrictive lung disease (small chest)	Restrictive lung disease (small chest)	Slightly reduced lung weight/body weight ratio, normal lung histology	Small chest
Other				
Dysmorphic craniofacial features	Midface retrusion, malar flattening, Shallow orbits, mandibular prognathia	Relative macrocephaly, triangular shaped facies, proptosis, downslanting palpebral fissures, mildly low- set ears	Low-set ears, micrognathia, high arched palate, flattened nares	Hypertelorism, low-set ears, flat nasal bridge
Congenital heart defect	Moderate secundum ASD	No	No	Narrow pulmonary valve, dilated pulmonary artery branches, overriding aorta, absent ductus arteriosus
Ophthalmology	Drooping eyelids (Lt>Rt); bilateral lagophthalmos, hyperopia	Blue sclera, drooping eye lids	N/A	N/A
Endocrine	Delayed puberty	None	N/A	N/A

#### Table 1. Clinical summary and KIF5B variants identified.

(Continued)

	Proband 1	Proband 2	Proband 3	Proband 4		
Development & cognitive function	Mild developmental delay (gross motor>fine>speech), normal cognitive function	Normal cognitive function	N/A	N/A		
Additional testing						
Chromosomal microarray analysis	Normal	Not done	Normal	Normal		
Variants in OI-related genes	Heterozygous VUS in <i>P3H1</i> (p. Asn491Ser), maternally inherited	COL1A1 and COL1A negative OI gene panel (34 genes) negative	None	None		
Biochemical collagen analysis	Normal migration (no overmodification) and normal 3-hydroxyproline analysis in fibroblasts	Not done	Not done	Not done		

#### Table 1. (Continued)

<u>Abbreviations</u>: ASD = atrial septal defect, BW = birth weight, IUGR = intrauterine growth restriction, POC = product of conception, N/A = not applicable, OI = osteogenesis imperfecta, SGA = small for gestational age at birth, VUS = variant of uncertain clinical significance.

https://doi.org/10.1371/journal.pgen.1011005.t001

adding wild type copies will suppress the phenotype, however, if the variant leads to a hyperactive gene product, then adding wild type copies will enhance the phenotype. The endogenous unc-116 gene maps to chromosome III (Fig 2E, green). Using recombination-mediated cassette exchange (RMCE) [29,30], we inserted an additional copy of the wild-type unc-116 gene into a safe harbor site on chromosome II (Fig 2E, blue). RNA-seq analysis confirmed that this extra copy was expressed at the same level as the endogenous gene (S3A Fig). Rescue studies demonstrated that two copies of unc-116(+) transgene were able to almost completely restore unc-116(del) mutants to wild-type level in body length and thrashing speed, indicating the transgene was fully functional (S3B-S3C Fig). Moreover, *unc-116(T90T)* control heterozygotes expressing two additional copies of the unc-116(+) transgene had the same body length and thrashed at the same speed as the wild-type animals indicating that expression of extra copies of the transgene did not adversely alter the phenotype (Fig 2F and 2G). In the unc-116 (T90I) variant heterozygotes, one additional copy of unc-116(+) transgene partially suppressed the body length defect, while two additional copies fully restored the body length to wild-type levels (Fig 2F). Similar rescuing effects on thrashing were observed with additional copies of the unc-116(+) transgene (Fig 2G). In summary, our data indicates that three copies of the wild-type allele were required to suppress one copy of the variant allele, providing support that the unc-116(T90I) variant produces a dominant negative gene product that interferes with UNC-116 wild type function.

#### The UNC-116 Thr90Ile variant impairs kinesin motor transport function

To directly test the effect of the T90I variant on UNC-116 kinesin motor function, we visualized mitochondrial cargo transport along microtubules in the anterior neurite of the ALM touch neurons using the transgene, *jsIs609 (mec-7p::mitoGFP)* (S1 Table) [31]. We employed time-lapse confocal imaging to generate a 2D kymograph to track and quantify mitochondrial movement [32]. The kymographs identified several stationary (Fig 3A, vertical lines, *red arrowheads*) and moving mitochondria particles (Fig 3A, diagonal lines, *green arrowheads*). The number of moving particles was significantly reduced in *unc-116(T90I)* heterozygotes compared to wild-type and *unc-116(T90T)* heterozygote controls (Fig 3B and S3 and S4 Movies). Moreover, these particles traveled shorter distances (Fig 3C) and moved at a slower speed



**Fig 2.** *unc-116(T90I)* is damaging and behaves as a dominant negative allele in *C. elegans.* (A) Images of wild-type (VC2010), *unc-116(T90T)* control edit, *unc-116(T90I)* variant edit, and *unc-116(del)* null worms. Homozygous *unc-116(T90I)* and *unc-116(del)* animals were L1-arrested and severely uncoordinated (Unc). *unc-116(del)* escaping adults were Unc and dumpy. Worms were grown for 7 days after egg-laying. Scale bar: 100 µm. (B) Developmental stages of animals scored 4 days after egg-laying (except for "Adult (7 days)" which were the total number of worms that grew to be adults after 7 days). *T90I#1* and *T90I#2* are two independent variant lines generated by CRISPR/Cas9 editing. Quantification of (C) body length and (D) thrashing speed of *unc-116(T90T#1)* and *unc-116(T90I#2)* heterozygotes. (E) Illustration showing the endogenous *unc-116* locus on chromosome III and the single-copy wildtype *unc-116* transgene integrated into a safe harbor locus on chromosome II. (F) Quantification of body length and (G) thrashing speed of *unc-116* transgenes. Statistical analysis was performed using the Kruskall-Wallis multiple comparison tests. ns: not significant, \*p<0.01, \*\*p<0.001, \*\*\*p<0.0001.

https://doi.org/10.1371/journal.pgen.1011005.g002

(Fig 3D) indicating that the kinesin motor activity was impaired in the UNC-116 T90I variant protein.

In summary, the experimental studies in *C. elegans* provide strong evidence for the UNC-116 T90I variant being damaging and causing defects in post-embryonic development, body length, locomotion, and kinesin motor function by acting as a dominant negative protein. To assess the effect of the *KIF5B* variant p.Thr87Ile on skeletal development, we used CRISPR/ Cas9 editing to generate a *Kif5b*<sup>T87I</sup> knock-in mouse model. However, no viable mice were identified by screening, likely due to early embryonic lethality.

## The *KIF5B* p.Thr87Ile variant is associated with an intracellular trafficking defect

To understand the cellular consequences of the recurrent *KIF5B* p.Thr87Ile variant we performed a microscopy study in primary patient fibroblasts. Electron microscopy on fibroblasts obtained from proband 1 showed prominent and dilated endoplasmic reticulum (ER) loops with abundant granular material, and multiple vacuoles containing cell breakdown products (**Fig 4A**). This pattern suggests a disruption of intracellular trafficking that is expected to affect the organization and function of the ER-Golgi complex. Consistently, immunofluorescent staining for the cis-Golgi matrix protein GM-130 in proband 1 fibroblasts showed altered Golgi distribution, with collapsed Golgi stacks (**Fig 4B**). Quantification in cells labeled with CellLight Golgi-GFP reagent showed decreased Golgi area as compared to control fibroblasts (**Fig 4C**).

## Primary cilia length is altered in cells expressing the *KIF5B* p.Thr87Ile variant

KIF5B has been implicated in the biogenesis of primary cilia and specifically was shown to regulate cilia length via its interaction with the BBSome complex [33]. The primary cilia were imaged in NIH3T3 cells overexpressing the human *KIF5B* p.Thr87Ile allele. Immunofluorescence labeling for acetylated tubulin demonstrated elongated primary cilia in cells expressing mutant *KIF5B* but not in cells overexpressing the wild type allele or in *Kif5b*-depleted cells (**Figs 5A, 5B** and **S4**). Similarly, the primary cilia were elongated in fibroblasts from proband 1 (**Fig 5B**).

#### mTOR signaling is downregulated in KIF5B mutant cells

Disruption of the Golgi-primary cilia axis can affect downstream intracellular signaling events [34-36]. To identify dysregulated signaling pathways, we performed differential gene expression analysis via mRNA sequencing in proband 1 fibroblasts as compared to Undiagnosed Diseases Network (UDN) control fibroblast transcriptome data (n = 131) [37]. The analysis





https://doi.org/10.1371/journal.pgen.1011005.g003

revealed reduced expression of multiple genes encoding ribosomal proteins. Ingenuity Pathway Analysis (IPA) of the gene expression profile suggested downregulation of mTOR signaling upstream of these transcriptional alterations (**Fig 6A**). Immunohistochemistry staining for phosphorylated ribosomal protein S6 confirmed a decrease in mTOR signaling *in situ*, in proband 1 bone biopsy (**Fig 6B**). Western blot in proband 1 fibroblasts showed reduced



Fig 4. Dilated endoplasmic reticulum and altered Golgi distribution in patient cells suggest an intracellular trafficking defect. (A) Electron microscopy of proband 1 fibroblasts showing dilated endoplasmic reticulum (red arrows) and multiple vacuoles containing cell breakdown material (yellow arrows). (B) Representative image showing altered Golgi distribution in patient (proband 1) fibroblast cells (right) as compared to healthy control (left). (C) Quantification of the Golgi circularity and Golgi area in patient (proband 1) fibroblasts. Left: Decreased circularity in patient cells, measured as the fraction of nucleus surrounded by Golgi stacks, on a scale of 0 (highly polarized) to 1 (radially distributed). Right: Reduced Golgi area in patient cells, measured as the total area occupied by Golgi stacks. Graphs summarize multiple single-cell measurements (Mann-Whitney U test, n = 321 control fibroblasts, n = 887 patient fibroblasts, \*\*P<0.001, \*\*\*p<0.0001).

https://doi.org/10.1371/journal.pgen.1011005.g004



**Fig 5.** The *KIF5B* **p.Thr87Ile variant is associated with elongated cilia.** (A) Representative image showing elongated cilia in NIH3T3 cells overexpressing *KIF5B*<sup>T87I/+</sup> (Mut, middle), compared to empty vector (EV, left) or wild type *KIF5B* (WT, right). (B) Left: Cilia length was not altered by KIF5B depletion (control siRNA n = 43, *Kif5b* siRNA n = 36). Middle: In contrast, overexpression of the mutant *KIF5B*<sup>T87I</sup> was associated with elongation of the cilia, as compared to cells transfected with the wild-type allele or with empty vector (EV n = 69, WT n = 70, Mut n = 51). Right: Cilia length was significantly increased in primary patient's fibroblasts (control human fibroblasts n = 30, proband 1 fibroblasts n = 28). T-test or one-way ANOVA, ns: not significant, \*p<0.01, \*\*\*p<0.0001.

https://doi.org/10.1371/journal.pgen.1011005.g005

phosphorylation of ribosomal protein S6, as well as decreased phosphorylation of AKT and mTOR proteins (S5 Fig). To potentially rescue this defect with a clinically translationally relevant intervention, we tested whether leucine amino acid supplementation could rescue this mTOR signaling defect. The essential amino acid leucine has been implicated in the regulation of mTOR signaling via mTORC1 activation [38–40]. As predicted, ribosomal protein S6 phosphorylation was restored after incubation with serum-containing media or by leucine supplementation (Fig 6C).

#### Discussion

Kinesins are molecular motors that transport mRNA, proteins, and subcellular organelles within cells by moving the cargo along oriented microtubules [1-3]. *KIF5B*, encoding the



Fig 6. The *KIF5B* p.Thr87Ile variant is associated with reduced mTOR signaling. (A) Differential expression of curated gene list representing mTOR pathway, based on RNA sequencing analysis from proband 1 fibroblasts compared to UDN control group (top: transcripts that are downregulated in patient, bottom: transcripts that are upregulated in patient). (B) Bone biopsy from proband 1 (Patient) compared to control. Periosteal surface stained positive for phospho-S6 in control bone (red arrowheads), but not in the patient's bone, which has immature and fibrotic bone matrix with hypercellularity that is typical of OI (Upper panel: Phospho-S6, bottom panel:  $2^{nd}$  antibody only, x10 scale bar = 20µm). (C) Left: Western blot showing reduced phosphorylated S6 ribosomal protein (pS6) in serum-starved patient fibroblasts (top panel) compared to control cells. Results normalized to total S6 (tS6) and tubulin ( $\alpha$ -tub). Refeeding with serum (middle panel), or leucine supplementation (bottom panel) reverses the phenotype in patient cells. Right: Quantification of immunoblot intensity using ImageJ (One-way ANOVA, n = 3 technical repeats, ns: not significant, \*\*\*p<0.0001).

https://doi.org/10.1371/journal.pgen.1011005.g006

ubiquitously expressed kinesin-1 heavy chain subunit, is essential for cell function and tissue development [10–14] and has been implicated in human disease [23,24]. We report here pathogenic variants in *KIF5B* in individuals with osteogenesis imperfecta. The variants that reside within the kinesin motor domain are predicted to interrupt nucleotide binding. Functional assessment of the recurrent p.Thr87Ile variant modeled in *C. elegans unc-116(T90I)* showed that the variant is damaging and disrupts post-embryonic development, body morphology, locomotion and kinesin motor activity via a dominant-negative mechanism. Studies in cells showed disrupted organization of the Golgi complex and altered cilia length, consistent with previous reports in *KIF5B*-mutated animal and cell culture models [12,24,33,41]. Importantly, downstream effect on the mTOR pathway was demonstrated in patient cells and bone tissue, highlighting a potential signaling defect that could be targeted for future therapy.

Kinesin-1 is an ATP-dependent motor protein [8], and the nucleotide-binding P-loop and switch loop I/II subdomains are critical to its activity. In vitro studies of key amino acid substitutions within the kinesin motor domain, including at Thr87, destabilized the nucleotidebinding pocket and favored an apo-kinesin (nucleotide-free) conformation that is strongly bound to microtubules [42-45]. The variants at Thr87, Gly90 and Thr195 are predicted to result in a catalytically inactive KIF5B protein due to destabilization of the active (ATP-bound) form (Fig 1E). In this scenario, heterozygous variants in the patients will likely have a dominant negative effect through dimerization of the catalytic-dead KIF5B with the wild type KIF5B. Of note, pathogenic variants affecting the homologous kinesin KIF1A have been proposed to act in a similar manner in autosomal dominant KIF1A-associated neurological disorder [46]. Consistent with this hypothesis, the unc-116(Thr90Ile) heterozygotes in C. elegans displayed dominant motility and body morphology phenotypes that were not observed in unc-116 deletion heterozygotes. Moreover, these phenotypes were suppressed by expression of additional copies of the wild type allele supporting a dominant negative mechanism (Fig 2). Similarly, the altered ciliary length in cells expressing mutant p.Thr87Ile KIF5B was not observed in *Kif5b*-depleted cells, nor in cells overexpressing the wild type *KIF5B* (Fig 5). Clustering of the KIF5B variants p.Thr87Ile, p.Gly90Ala and p.Thr195Lys (reported here), as well as p.Lys91Arg and p.Gly234Val [23] within the kinesin motor domain suggests a genotypephenotype correlation, compared to previously reported variants residing outside this domain that are not associated with skeletal findings [24] (Fig 1D).

Kinesins function in vesicle trafficking, including anterograde transport from the ER to the Golgi complex, trafficking within the Golgi cisternae, and transport between the Golgi, plasma membrane and other intracellular organelles [1,5,7,41,47]. Additionally, mouse Kif5b loss of function is known to disrupt mitochondrial transport [10,12]. Direct assessment of kinesin motor activity in C. elegans unc-116(Thr90Ile) heterozygotes showed reduced mitochondrial trafficking along microtubules in ALM neurites (Fig 3) providing support for the deleterious nature of the human p.Thr87Ile variant on kinesin function. Altered organization of the Golgi stacks was previously described in mouse Kif5b-deficient myogenic and chondrogenic cells [12,13], and in fibroblasts derived from individuals with KIF5B-related disorder [24]. In proband 1 fibroblasts, electron microscopy and immunofluorescence imaging demonstrated altered ER morphology, multiple vacuoles, and abnormal distribution of the Golgi complex (Fig 4). This pattern correlates with disruption of intracellular trafficking. Vesicular trafficking defects have been implicated in low bone mass phenotypes and bone fragility that are at least partly attributed to failure in collagen secretion [48-50]. Interestingly, impaired secretion of type II collagen has been demonstrated in *Kif5b*-deficient chondrocytes [14,51]. In proband 1 fibroblasts, type I collagen electrophoresis did not suggest a qualitative or quantitative collagen defect (S1 Fig). More research is needed to determine whether impaired collagen secretion is contributing to the phenotype in KIF5B-related osteogenesis imperfecta.

We also studied the effect of mutant KIF5B on cilia biogenesis. Overexpression of the wild type *KIF5B*, or its depletion by siRNA, did not significantly affect the cilia length. In contrast, expression of the mutant KIF5B<sup>T87I</sup> was associated with elongated cilia, as has been observed in the proband 1 cells (**Fig 5**). This can be potentially explained by the direct effect of KIF5B on ciliogenesis via its interaction with CCDC28B at the ciliary basal body [33]. Alternatively, since primary cilia assembly and function depend on adequate protein transport through the Golgi, dysregulated cilia length may be secondary to the disruption of the secretory pathway in KIF5B<sup>T871</sup> mutant cells [52,53].

The secretory pathway initiates and propagates signal transduction, including the mitogenactivated protein kinase (MAPK) and mammalian target of rapamycin (mTOR) signaling pathways, via proteins residing in the ER and Golgi membranes [54]. In doing so, it not only regulates processes like cell proliferation and differentiation, but also modulates its own organization and function [54,55]. Our data demonstrated reduced mTOR signaling in patient fibroblasts and bone tissue (Fig 6). Santos-Ledo et al. had previously reported decreased mTOR phosphorylation and reduced phosphorylated ribosomal protein S6 in chondrocytes from kif5b-deficient zebrafish [14]. *KIF5B* plays an important role in lysosome positioning and in maintaining lysosome homeostasis, and KIF5B-depleted cells exhibit dysregulated mTOR signaling due to altered activation of the lysosome-associated mTORC1/mTORC2 complexes [56-59]. The Golgi apparatus also serves as a functional hub for mTOR proteins, and loss of Golgi integrity resulted in altered mTOR activity and induction of autophagy [60]. Thus, reduced mTOR signaling in KIF5B patients may be secondary to the disruption of organelle distribution and function in the setting of an intracellular trafficking defect. The dysregulation of mTOR signaling may also be related to altered primary cilia, or conversely may explain the abnormal ciliogenesis in cells expressing mutant KIF5B [61]. We have shown that loss of phosphorylated S6 in proband 1 cells can be partially rescued by supplementation with leucine, a potent stimulator of mTOR signaling (Fig 6). Leucine supplementation has been previously proposed as a therapeutic approach in other genetic disorders, including Roberts syndrome and Diamond-Blackfan anemia, that are also characterized by dysregulation of mTOR signaling, protein translation and autophagy [62–64]. Thus, our findings may have future therapeutic implications for patients with KIF5B-related disorder.

Pathogenic variants in *KIF5B* were previously reported in individuals with kyphomelic dysplasia, a skeletal dysplasia characterized by short stature, narrow thorax, bowing of the long bones and dysmorphic craniofacial features [23]. Probands 1–4 reported here were clinically diagnosed with osteogenesis imperfecta based on their history of osteopenia, multiple fractures, and bone deformities. The phenotypic overlap between reported patients, including short stature, bowed long bones, small chest, osteoporosis, fractures, and extra-skeletal manifestations (such as ptosis and hearing loss), suggests that these individuals may in fact be part of a clinical spectrum (Table 1 and [23]). Identification of additional individuals with *KIF5B* variants will allow for better characterization of the phenotype in *KIF5B*-related disorder.

One limitation of this study is the inability to model the skeletal dysplasia *in vivo*. We were unsuccessful in generating a *Kif5b*<sup>T871</sup> knock-in mouse model, likely due to early embryonic lethality, as has been reported in the *Kif5b* knockout mice [10]. The specific role of *KIF5B* in bone is not well-understood. An intracellular trafficking defect could potentially lead to failure in collagen secretion, and may be associated with ER stress, Golgi dysfunction, autophagy, and altered signaling, ultimately affecting osteo-chondroprogenitor cells' differentiation or extracellular matrix composition. KIF5B deficiency was associated with a defect in cartilage development and skeletal mineralization in zebrafish and in the conditional Col2a1-Cre mouse model [13,14]. We had shown down regulation of the mTOR signaling pathway, which plays a critical role in bone development. Notably, reduced mTOR signaling has been previously

implicated in mouse models of osteogenesis imperfecta [65–67]. Together, this provides a mechanistic rationale for the abnormal skeletal phenotype in the patients.

In summary, we have identified heterozygous variants in *KIF5B* as a novel cause of osteogenesis imperfecta with a dominant negative mechanism due to an intracellular trafficking defect and downregulation of mTOR signaling pathway. This study emphasizes the critical role of KIF5B and intracellular trafficking in maintaining skeletal homeostasis.

#### Methods

#### **Ethics statement**

For human studies, written informed consent was obtained from participants, or from their parent/legal guardian, as part of the Undiagnosed Diseases Network (UDN) for probands 1 and 2, according to the institutional review board of the National Human Genome Research Institute (15HG0130). Proband 1 was also enrolled to a research protocol for skeletal dysplasias that was approved by the Baylor College of Medicine Institutional Review Board (H-25722). Probands 3 and 4 were ascertained by diagnostic testing procedures, and their legal guardians gave clinical informed consent for testing. This was obtained using standard forms at local site by the responsible referring physicians.

#### Exome sequencing and identification of KIF5B variants

Proband 1 had research trio exome sequencing performed at the Baylor College of Medicine Human Genome Sequencing Center (BCM-HGSC). Briefly, library was constructed as described in the BCM-HGSC protocol and hybridized to the HGSC VCRome 2.1 design (42Mb NimbleGen, Cat. No. 06266380001) [68]. Sequencing was performed in paired-end mode using the Illumina HiSeq 2000 platform. With a sequencing yield of 9.1 Gb, the samples achieved an average coverage of 106x and 93.4% of the targeted exome bases were covered to a depth of 20X or greater. Illumina sequence analysis was performed using the HGSC Mercury analysis pipeline [69,70] which moves data through various analysis tools from the initial sequence generation on the instrument to annotated variant calls. In parallel to the exome workflow an Illumina Infinium Exome-24 v1.1 array was generated for a final quality assessment. This included orthogonal confirmation of sample identity and purity using the Error Rate In Sequencing (ERIS) pipeline developed at the HGSC. A successfully sequenced sample met quality control metrics of ERIS SNP array concordance (>90%) and ERIS average contamination rate (<5%). Variants were reviewed using Codified Genomics software (https:// www.codifiedgenomics.com). We prioritized variants that were de novo, biallelic, and Xlinked with a minor allele frequency of <0.01% within gnomAD (https://gnomad. broadinstitute.org/) [25] and our internal database. Proband 2 had research whole genome sequencing performed at Baylor Genetics. In brief, genomic DNA was fragmented and indexed to prepare individual library, followed the PCR free 550-bp insert size protocol by the KAPA Hyper Prep kit. The library was subjected to sequence analysis on Illumina NovaSeq 6000 platform for 150 bp paired end reads with average sequenced coverage over the genome > 40X. Data analysis and interpretation were performed by the Baylor Genetics analytics pipeline. Probands 3 and 4 underwent diagnostic trio exome sequencing in the Exeter Genomics Laboratory (Royal Devon University Healthcare NHS Foundation Trust, Exeter, UK). Libraries were prepared and targets captured using the Twist Exome kit (Twist Bioscience, South San Francisco, CA, USA) according to the manufacturer's instructions. Sequencing was carried out on an Illumina NextSeq550 (Illumina, San Diego, CA, USA) using 150-bp paired end reads. All libraries met minimum quality requirements of mean depth 80X or greater and >95% of targeted bases covered to a depth of 20X or greater. Variants were called

in human genome build GRCh37 and annotated using Alamut Batch software (Interactive Software, Rouen, France).

#### **RNA** sequencing

Messenger RNA from fibroblasts of proband 1 and UDN controls was extracted, quantified, and processed, prior to library preparation using the Illumina TruSeq kit (Illumina, SD) according to manufacturer's instructions. Samples were subsequently multiplexed and subject to 150bp paired-end sequencing on the Illumina HiSeq2000 platform. Resulting raw bcl reads were converted to FASTQ prior to being processed using a pipeline adapted from the Genotype-Tissue Expression (GTEX) Consortium [71]. Briefly, reads were aligned to the GRCh37 human reference using Spliced Transcript Alignment to a Reference (STAR) [72], followed by gene-based quantification with RNA-Seq by Expectation Maximization (RSEM) [73] and alignment visualization in Integrative Genomics Viewer (IGV) [74]. For gene expression analysis, FPKM values (fragments per kilobase of transcript per million fragments mapped) in the proband RNA-seq data were compared to the mean FPKM value from pooled control RNA-seq data. The p value, mean and standard deviation were calculated based on the normal distribution of FPKM values among samples. Z score values were calculated as (observed FPKM-Mean)/standard deviation. The significantly differential expressed genes (fold change >2 and p value <0.001) were then uploaded into IPA (Qiagen) for upstream regulator prediction.

#### Protein 3D structure analysis

Structural impact of variants was assessed in structures of KIF5B bound to either ADP or ATP (Protein Data Bank structures 1mkj and 3j8y, respectively). Variants were introduced by *in silico* mutagenesis using the FoldX modeling suite [75]. Structures were visualized and images generated using PyMOL (PyMOL Molecular Graphics System, Version 2.0, Schrödinger LLC; New York, NY, USA).

#### **Electron microscopy**

Fibroblast cells were pelleted and fixed in 4% paraformaldehyde, washed, and dehydrated in a graded ethanol series from 30–100%. The samples were then washed in propylene oxide and embedded in Spurrs epoxy. Ultrathin sections were stained in uranyl acetate followed by Reynolds lead citrate and examined using a FEI Tecnai G2 TEM.

#### Cell culture

All cells were cultured at 37 °C and 5% CO<sub>2</sub> in a humid environment. Primary human fibroblasts, or NIH3T3 cells were cultured in growth medium (alpha MEM supplemented with 10% fetal bovine serum, 1% glutamine and 1% penicillin/streptomycin). NIH3T3 cells were transiently transfected with either wild type *KIF5B*, mutant *KIF5B*<sup>T871</sup> (GenScript GenEZ ORF clone in pcDNA3.1 flag-tag) or empty vector (pcDNA3.1) using Lipofectamine 2000 reagent (Thermofisher). To knockdown *Kif5b* in NIH3T3 cells, cells were transfected with ON-TAR-GETplus Mouse *Kif5b* siRNA SMARTPool (Horizon) using Lipofectamine RNAiMax (Thermofisher). Total cell lysates were analyzed by western blot and real-time PCR to quantify KIF5B level. For measurements of cilia length, cells were subject to overnight starvation in growth media containing 0.5% fetal bovine serum. For studying mTOR pathway, cells were starved in serum-free media for 6 hours prior to sample collection. For rescue experiments, L-Leucine (AthenaES, 5mM) or fetal bovine serum (10%) were added to the serum-free media for 30 minutes prior to sample collection.

#### Type I collagen analysis

Primary human fibroblasts were cultured overnight in serum-free alpha MEM containing 50 uM ascorbic acid (Sigma) to stimulate collagen secretion. Media was harvested the next day, and protein was extracted by heat denaturation in SDS-PAGE sample buffer. Type I collagen  $\alpha$ -chains were resolved on 6% SDS-PAGE gel, then processed by trypsin in-gel digestion [76]. Digested fractions were analyzed for hydroxylation of proline residues by Liquid Chromatography/Mass spectrometry (LC/MS, Thermo Scientific).

#### Protein extraction and western blot

Protein was extracted from whole cell lysates in RIPA buffer. The protein concentration was measured using a BCA kit (ThermoFisher Scientific). For western blot analysis, 10–20 ug of total protein was loaded on 4–15% SDS-PAGE gradient gel (Bio-Rad) and transferred to PVDF membrane (Millipore Sigma). Membrane was blocked in 5% skim milk and incubated overnight at 4°C with primary antibody. The following primary antibodies were used: anti- DDDDK tag (Abcam #ab1257), anti- phospho-S6 ribosomal protein (Ser235/236) (Cell Signaling #2211), anti-S6 ribosomal protein (Cell Signaling #2217), anti-AKT (Cell Signaling #4691), anti-phospho AKT (Cell Signaling #9271), anti-mTOR (Cell Signaling #2983), anti-phospho mTOR (Cell Signaling #5536), and anti-alpha Tubulin (Sigma, T5168). Signal was captured via ChemiDoc imager (Bio Rad).

#### RNA extraction and real-time qPCR

Total RNA was extracted from cell lysates in Trizol reagent (Ambion). The iScript cDNA synthesis kit (Bio-Rad) was used to synthesize cDNA. Real-time qPCR was performed on LightCycler instrument (Roche) using FastStart Essential DNA Master reagent (Roche) with *B2m* (beta-2 microglubulin) as internal control.

#### Immunofluorescence microscopy

Cells were cultured in chamber slides, fixed in 4% formaldehyde, blocked with 5% serum in PBS/0.3%Triton X-100, and incubated overnight at 4°C with primary antibody. Primary antibodies used in this study were: anti-GM-130 (Cell Signaling #12480) and rabbit anti-acetyl- $\alpha$ -Tubulin (Cell Signaling #5335). Secondary antibody was donkey anti-rabbit IgG (H+L), Alexa Fluor 594 (Thermofisher #A-21207). Slides were mounted using Prolonged Gold Antifade mounting medium with DAPI (Thermofisher). Acquisition of images was done with the assistance of the Baylor College of Medicine Integrated Microscopy Core, using Cytivia DVLive epifluorescence deconvolution microscope. Images were captured on a sCMOS camera taking z-stacks with 0.2µm optical sections. Cilia length was measured using ImageJ.

#### Golgi imaging in human primary fibroblasts

Quantitative imaging of Golgi structure and distribution was performed in human skin fibroblasts from proband 1 or control cells. Fibroblasts were cultured in chamber slides, and Golgi complex was labeled by CellLight Golgi-GFP BacMan 2.0 reagent per manufacturer's protocol (ThermoFisher Scientific). Staining with Hoechst was used to label nuclei. Acquisition and analysis of images was done with the assistance of the Baylor College of Medicine Integrated Microscopy Core, using Vala IC200 high throughput microscope. Nuclear size and intensity, as well as number of Golgi structures, area, mean pixel intensity, and maximum length were measured. A measurement scale of 0 to 1 was used to identify the relative radial distribution of each Golgi. Golgi circularity was scored on a scale of 0 to 1, with 0 representing a Golgi that is polarized to one region, and a score of 1 representing a fully dispersed Golgi around the nucleus.

#### Immunohistochemistry in bone biopsy

Bone samples from proband 1 and unaffected control were obtained when subjects were undergoing skeletal surgery for a medical indication. The fragments of bone removed during surgery, which otherwise would have been discarded, were processed for histology. In brief, bone samples were fixed in 4% paraformaldehyde for 48 hours and decalcified in 10% EDTA (Millipore Sigma) for 14 days at 4°C. Paraffin embedded tissue was sectioned, deparaffinized, and incubated at 37°C with 0.05% trypsin for antigen retrieval (Abcam ab970), followed by 3% hydrogen peroxide treatment. After blocking with 5% normal goat serum, sections were incubated with primary antibody anti-phospho-S6 ribosomal protein (Ser235/236) (Cell Signaling #2211) overnight at 4°C. Anti-rabbit secondary antibody and DAB staining (Vector Laboratories, Vectastain Elite ABC-HRP Kit #PK-6101) were applied per manufacturer's instructions. Slides were stained with Hematoxylin and mounted using Cytoseal XYL xylene-based mounting medium (Thermo Scientific). Images were taken with a light microscope (Axioplan 2, Zeiss).

#### Modeling in C. elegans

*C. elegans* culture and strain information. *C. elegans* strains were cultured on standard nematode growth media (NGM) seeded with *Escherichia coli* OP50 and grown at 20°C. Strains used in this paper include *jsIs609* [*mec7p::mtGFP* + *lin-15*(+)] X and VC2010. *unc-116* variant edits, control edits, null deletion and single copy integration strains generated in this study are described below. A full list of strains used in this study is in <u>S1 Table</u>. Some strains were provided by the CGC, which is funded by NIH Office of Research Infrastructure Programs (P40 OD010440).

C. elegans genome editing. unc-116 edits were generated as previously described [77]. In brief, VC2010 adults were injected with Cas9 (IDT 1081060), tracrRNA (IDT 1072533), crRNA targeting sequence GCATATGGACAAACATCTTC (IDT CRISPR-Cas9 sgRNA), single-stranded DNA oligonucleotide repair template (IDT), and *dpy-10* crRNA and repair template (the co-conversion marker) [78,79]. The variant repair template includes the proband edit and synonymous changes, which destroy Cas9 re-targeting and create a TspRI restriction site for genotyping. The control repair template with the synonymous changes, but not the proband variant, was used to generate control edit lines to verify that the synonymous changes do not contribute to the observed phenotype in the variant edit animals. Three T90I variant edit lines and two T90T control edit lines were initially generated. T90I#1, T90I#2 and T90T#1 were Sanger sequenced verified and outcrossed twice to remove any potential background mutations. Null deletion allele unc-116(gk5722udn86) is derived from unc-116(gk5722[loxP + myo-2p::GFP::unc-54 3' UTR + rps-27p::neoR::unc-54 3' UTR + loxP]). Selection cassette (myo-2p::GFP::unc-54 3' UTR + rps-27p::neoR::unc-54 3' UTR) in unc-116(gk5722) was removed by injecting a plasmid pDD104 expressing Cre recombinase (a gift from Bob Goldstein, Addgene plasmid # 47551) as described previously [80]. The single-copy unc-116(+) transgene was integrated at the Mos1 transposon insertion site ttTi5605 on chromosome II (II: 0.77cM) using recombination-mediated cassette exchange (RMCE) [29,30]. The entire unc-116 genomic locus was amplified from N2 genomic DNA using oligonucleotides 6932 and 6947, digested with MluI and SbfI and inserted into pLF3FShC. The resulting plasmid NM3933 was injected at 50 ng/µl into NM5340 jsSi1579 [loxP rpl-28p FRT GFP::his-58 FRT3] II; unc-119(ed3) III; bqSi711 [mex-5pFLP sl2 mNeon Green] IV. Three independent integrants were obtained as previously described [81]. After outcrossed twice with VC2010 to remove unc-119(ed3) III and bqSi711 [mex-5pFLP sl2 mNeon Green] IV, three single copy unc-116(+) transgene lines were generated.

The unc-116 transgene RNA-seq analysis. To facilitate the comparison of mRNA level between the unc-116 wild type and transgene locus in RNA-seq analysis, we have introduced four synonymous substitutions in exon 3 at the endogenous locus to generate unc-116(udn42). We conducted the RNA-seq analysis in triplicates to compare the mRNA level/read counts between the unc-116 wild type locus (chromosome III) and single copy transgene locus on chromosome II, using the strain UDN100221. The UDN100221 harbors the unc-116 transgene allele *jsSi1579 jsSi1606* and the edited allele *und42* at the endogenous locus [e.g., *unc-116* (udn42)]. In brief, 1-day old young adults were collected to isolate total RNA with Trizol reagent following the manufacturer's instruction. Total RNA was further purified with DNase digestion followed by column extraction. The following Next Generation Sequencing (NGS) DNA library preparation and bioinformatic analysis were conducted by Genome Technology Access Center: ~10 µg of total RNA each was used to enrich mRNA before DNA library preparation, using Oligo-dT beads (mRNA Direct kit, Life Technologies). mRNA was first fragmented in reverse transcriptase buffer by incubation at 94°C for 8 minutes. mRNA was then reverse transcribed to yield cDNA using the SuperScript III RT enzyme (Life Technologies) with random hexamers. A second strand reaction was performed to yield double-strand cDNA. cDNA was blunt ended, had an A base added to the 3' ends, and then Illumina sequencing adapters were ligated to the ends. Ligated fragments were then PCR amplified for 12–15 cycles, using primers to integrate unique dual index tags for the different biological replicates. Fragments were sequenced on an Illumina NovaSeq-6000 using paired-end reads extending 150 bases. The resulting reads were then quasi-aligned and quantitated against the Ensembl WBcel235.90 transcriptome with copies of the wild type and transgenic unc-116 using the expectation-maximization algorithm in Salmon [82] to generate allele-specific expression in transcripts per million (TPM). The results indicate that the mRNA level of the single copy unc-116 transgene is not significantly different from the endogenous locus, and thus can be considered as supplying an equivalent gene dose as the wild type unc-116 gene.

*C. elegans* body length and swimming analysis. Body length and swimming speed were measured using the WormLab 2019.1.1 (MBF Bioscience) [81]. Briefly, about 30 late L4 stage worms per strain were transferred onto seeded NGM plates 24 hours before the assay. Meanwhile, 100 mm NGM plates were seeded with 50  $\mu$ L 10x concentrated OP50, spreading onto the plate surface evenly, and allowed to dry for 1 day. On the day of assay, 10 animals were transferred to one newly made assay plate, recovering for 20 minutes, and recorded for 2 minutes of worm crawling. After that, 1mL PBS was added into the plate, waiting for 30 seconds for worms to adapt to swimming, and then 2 minutes of worm swimming video was captured. Three plates per strain were recorded per trial and three independent trials were performed for each strain, totaling about 90 animals per strain. Videos were analyzed using WormLab software. When multiple tracks were generated by one worm, tracks were manually joined when possible, otherwise the longer duration track was used. Any worms that recorded less than 15 seconds were censored from analysis.

**Generation of variant heterozygotes.** To obtain heterozygotes, L4 stage animals were transferred to *fem-1* RNAi plates to generate females. Adult females, identified by presence of stacked oocytes but not embryos, were selected to cross with males of appropriate genotype. Therefore, the yielding offspring were all cross progeny. Specifically, for *unc-116/qC1*::GFP, females were crossed with VC2010 males and heterozygote progeny without pharyngeal GFP were selected for phenotyping.

**Image acquisition and analysis.** For imaging UNC-116 mediated transport of mitochondria along axonal microtubules, L4 worms with GFP tagged mitochondria in mechanosensory neurons were immobilized with 10 mM levamisole, mounted on 2.5% agarose pads, and sealed with Vaseline. 5 minutes time-lapse images (1fps) were acquired using 63x objective on a spinning disk confocal microscope (PerkinElmer). Time-lapse images were converted into 2D kymographs by KymoResliceWide plugin in ImageJ. Images were aligned with StackReg Plugin in ImageJ where worm movement was detected during imaging. 100  $\mu$ m long neurite regions, starting from ALM soma, were used for quantification of mitochondria moving particles. Particles with more than 3  $\mu$ m displacement were counted as moving particles. Anterograde moving particles were defined by particles moving away from the cell body. Net run length was calculated as the distance between the starting position and the stop position. For combined segmental velocity, each trajectory was manually tracked using segmented line tool and processed automatedly by KymoAnalyzer [32] in ImageJ. For images in Fig 2, worms were imaged with Leica M165 FC Fluorescent Stereo Microscope. For confocal images in Fig 3, "Straighten" function of ImageJ was used to straighten curved ALM neurites.

**Statistical analysis.** Statistical analysis was performed using GraphPad Prism (Version 8.0.2). Student's t-test or Mann-Whitney test were used for comparing differences between two groups, while one-way ANOVA or Kruskal-Wallis test followed by Dunn's multiple comparison was performed for comparing three or more groups. For *C. elegans* modeling: Data were shown as median value, except Figs <u>2B and S3A</u> in which mean and SD were shown. For cell studies: Data were shown as mean value and SD. Statistical significance was determined as p value that is equal to, or less than 0.01.

#### Supporting information

**S1 Movie. Control** *C. elegans* (WT, VC2010) swimming. (AVI)

**S2 Movie.** *unc-116(T90I)* variant (Het) swimming. (AVI)

**S3** Movie. *unc-116(T90T)* control edit mitochondrial transport. (AVI)

**S4 Movie.** *unc-116(T90I)* variant (Het) mitochondrial transport. (AVI)

**S1 Data. Supplemental data file, RNA sequencing data.** (XLSX)

S2 Data. Supplemental data file, numerical data (cell studies). (XLSX)

**S3 Data. Supplemental data file, numerical data (***C.elegans* studies). (XLSX)

**S1 Fig. Collagen post-translational modification is not altered in Proband 1.** Type I collagen isolated from patient fibroblasts showed no difference in electrophoretic migration pattern between patient and control. Mass spectrometry analysis showed no difference in hydroxylation pattern of 3-Hydroxyproline at P986 between patient and control. (PDF)

**S2 Fig.** *KIF5B* P-loop nucleotide binding domain is highly conserved between species. Multiple sequence alignment of the N-terminal part of the motor domain of *KIF5B* and orthologs in other organisms. The highly conserved P-loop region of the ATP-binding domain is highlighted with a green box. The KIF5B p.Thr87Ile variant in proband 1 (red arrowhead) and corresponding *C. elegans* (worm) *unc-116* Thr90 residue (blue arrowhead) are shown. (PDF)

**S3 Fig. Single copy transgene of** *unc-116(+)* **is functional.** (A) RNA sequencing transcripts per million (TPM) of endogenous *unc-116* and *unc-116* transgene on chromosome *II*. Rescue of body length (B) and thrashing speed (C) defects were demonstrated upon introduction of two additional copies of *unc-116(+)* transgene into *unc-116 (del)*. The difference in length between wild type and *unc-116(del)* with 2 extra copies of *unc-116(+)* is less than 2% and while statistically different is not considered biologically significant. Student t-test was used for comparison; ns: not significant, \*\*p<0.001. (PDF)

**S4 Fig.** *KIF5B* overexpression and knockdown in NIH3T3 cells. (A) western blot analysis (left) and real time RT-PCR (right) showing overexpression of Flag-tagged wild type and mutant *KIF5B* in transiently transfected NIH3T3 cells (EV- empty vector, WT- wild type KIF5B, MUT- KIF5B<sup>T871</sup>). (B) Real time RT-PCR showing efficient knock-down of *Kif5b* in NIH3T3 cells using siRNA. ns: not significant, \*p<0.01, \*\*p<0.001, \*\*\*p<0.0001. (PDF)

**S5 Fig. Reduced mTOR signaling in patient fibroblasts.** Western blot showing reduced phosphorylated AKT (top panel) and mTOR (bottom panel) in serum-starved patient fibroblasts.

(PDF)

**S1 Table.** List of *C. elegans* strains used in this study. (PDF)

**S2 Table.** *unc-116* genome editing reagents. (PDF)

**S1** Acknowledgement. List of members of the UDN. (PDF)

#### Acknowledgments

The authors would like to thank the participants and their families for participation in the study. Some strains were provided by the Caenorhabditis Genetics Center (CGC), which is funded by NIH Office of Research Infrastructure Programs (P40 OD010440). Cell imaging for this project was supported by the Integrated Microscopy Core at Baylor College of Medicine and the Center for Advanced Microscopy and Image Informatics (CAMII) with funding from NIH (DK56338, CA125123, ES030285, S10OD030414), and CPRIT (RP150578, RP170719), and the assistance of Fabio Stossi and Hannah Johnson.

#### **Author Contributions**

- **Conceptualization:** Ronit Marom, Bo Zhang, Gary A. Silverman, Tim Schedl, Stephen C. Pak, Brendan H. Lee.
- **Data curation:** Lindsay C. Burrage, David R. Murdock, Hongzheng Dai, Shalini N. Jhangiani, Donna M. Muzny, Richard A. Gibbs, Richard Caswell, Karen Stals.
- Formal analysis: Lindsay C. Burrage, David R. Murdock, Hongzheng Dai, Shalini N. Jhangiani, Donna M. Muzny, Richard A. Gibbs, Richard Caswell, Karen Stals.

**Investigation:** Ronit Marom, Bo Zhang, Megan E. Washington, I-Wen Song, Vittoria C. Rossi, Ava S. Berrier, Anika Lindsey, Jacob Lesinski, Michael L. Nonet, Jian Chen, V. Reid

Sutton, M. John Hicks, MaryAnn Weis, Carrie Pottinger, Deirdre Cilliers, David Eyre, Deborah Krakow.

- Methodology: Ronit Marom, Bo Zhang, Megan E. Washington, I-Wen Song, Vittoria C. Rossi, Ava S. Berrier, Anika Lindsey, Jacob Lesinski, Michael L. Nonet, Jian Chen, M. John Hicks, MaryAnn Weis, David Eyre.
- Project administration: Dustin Baldridge, Jill A. Rosenfeld, Alyssa A. Tran.
- Supervision: Gary A. Silverman, Tim Schedl, Stephen C. Pak, Brendan H. Lee.
- Writing original draft: Ronit Marom, Bo Zhang, Tim Schedl, Stephen C. Pak, Brendan H. Lee.
- Writing review & editing: Ronit Marom, Bo Zhang, Megan E. Washington, I-Wen Song, Lindsay C. Burrage, Vittoria C. Rossi, Ava S. Berrier, Anika Lindsey, Jacob Lesinski, Michael L. Nonet, Jian Chen, Dustin Baldridge, Gary A. Silverman, V. Reid Sutton, Jill A. Rosenfeld, Alyssa A. Tran, M. John Hicks, David R. Murdock, Hongzheng Dai, MaryAnn Weis, Shalini N. Jhangiani, Donna M. Muzny, Richard A. Gibbs, Richard Caswell, Carrie Pottinger, Deirdre Cilliers, Karen Stals, David Eyre, Deborah Krakow, Tim Schedl, Stephen C. Pak, Brendan H. Lee.

#### References

- Ali I, Yang WC. The functions of kinesin and kinesin-related proteins in eukaryotes. Cell Adh Migr. 2020; 14(1):139–52. Epub 2020/08/28. https://doi.org/10.1080/19336918.2020.1810939 PMID: 32842864; PubMed Central PMCID: PMC7513868.
- Konjikusic MJ, Gray RS, Wallingford JB. The developmental biology of kinesins. Dev Biol. 2021; 469:26–36. Epub 2020/09/23. https://doi.org/10.1016/j.ydbio.2020.09.009 PMID: 32961118.
- Hirokawa N, Noda Y, Tanaka Y, Niwa S. Kinesin superfamily motor proteins and intracellular transport. Nat Rev Mol Cell Biol. 2009; 10(10):682–96. Epub 2009/09/24. https://doi.org/10.1038/nrm2774 PMID: 19773780.
- Cassella L, Ephrussi A. Subcellular spatial transcriptomics identifies three mechanistically different classes of localizing RNAs. Nat Commun. 2022; 13(1):6355. Epub 2022/10/27. https://doi.org/10. 1038/s41467-022-34004-2 PMID: 36289223; PubMed Central PMCID: PMC9606379.
- Serra-Marques A, Martin M, Katrukha EA, Grigoriev I, Peeters CA, Liu Q, et al. Concerted action of kinesins KIF5B and KIF13B promotes efficient secretory vesicle transport to microtubule plus ends. Elife. 2020; 9. Epub 2020/11/12. https://doi.org/10.7554/eLife.61302 PMID: 33174839; PubMed Central PMCID: PMC7710357.
- Cho KI, Cai Y, Yi H, Yeh A, Aslanukov A, Ferreira PA. Association of the kinesin-binding domain of RanBP2 to KIF5B and KIF5C determines mitochondria localization and function. Traffic. 2007; 8 (12):1722–35. Epub 2007/09/25. https://doi.org/10.1111/j.1600-0854.2007.00647.x PMID: 17887960.
- Pu J, Guardia CM, Keren-Kaplan T, Bonifacino JS. Mechanisms and functions of lysosome positioning. J Cell Sci. 2016; 129(23):4329–39. Epub 2016/11/02. https://doi.org/10.1242/jcs.196287 PMID: 27799357; PubMed Central PMCID: PMC5201012.
- Hirokawa N, Noda Y. Intracellular transport and kinesin superfamily proteins, KIFs: structure, function, and dynamics. Physiol Rev. 2008; 88(3):1089–118. Epub 2008/07/16. <u>https://doi.org/10.1152/</u> physrev.00023.2007 PMID: 18626067.
- Yip YY, Pernigo S, Sanger A, Xu M, Parsons M, Steiner RA, et al. The light chains of kinesin-1 are autoinhibited. Proc Natl Acad Sci U S A. 2016; 113(9):2418–23. Epub 2016/02/18. https://doi.org/10. 1073/pnas.1520817113 PMID: 26884162; PubMed Central PMCID: PMC4780624.
- Tanaka Y, Kanai Y, Okada Y, Nonaka S, Takeda S, Harada A, et al. Targeted disruption of mouse conventional kinesin heavy chain, kif5B, results in abnormal perinuclear clustering of mitochondria. Cell. 1998; 93(7):1147–58. Epub 1998/07/10. https://doi.org/10.1016/s0092-8674(00)81459-2 PMID: 9657148.
- Cui J, Wang Z, Cheng Q, Lin R, Zhang XM, Leung PS, et al. Targeted inactivation of kinesin-1 in pancreatic beta-cells in vivo leads to insulin secretory deficiency. Diabetes. 2011; 60(1):320–30. Epub 2010/09/28. https://doi.org/10.2337/db09-1078 PMID: 20870970; PubMed Central PMCID: PMC3012189.

- Wang Z, Cui J, Wong WM, Li X, Xue W, Lin R, et al. Kif5b controls the localization of myofibril components for their assembly and linkage to the myotendinous junctions. Development. 2013; 140(3):617–26. Epub 2013/01/08. https://doi.org/10.1242/dev.085969 PMID: 23293293.
- Gan H, Xue W, Gao Y, Zhu G, Chan D, Cheah KSE, et al. KIF5B modulates central spindle organization in late-stage cytokinesis in chondrocytes. Cell Biosci. 2019; 9:85. Epub 2019/10/23. https://doi. org/10.1186/s13578-019-0344-5 PMID: 31636894; PubMed Central PMCID: PMC6794761.
- Santos-Ledo A, Garcia-Macia M, Campbell PD, Gronska M, Marlow FL. Kinesin-1 promotes chondrocyte maintenance during skeletal morphogenesis. PLoS Genet. 2017; 13(7):e1006918. Epub 2017/ 07/18. https://doi.org/10.1371/journal.pgen.1006918 PMID: 28715414; PubMed Central PMCID: PMC5536392.
- Kalantari S, Filges I. 'Kinesinopathies': emerging role of the kinesin family member genes in birth defects. J Med Genet. 2020; 57(12):797–807. Epub 2020/05/21. https://doi.org/10.1136/jmedgenet-2019-106769 PMID: 32430361; PubMed Central PMCID: PMC7691813.
- Anazawa Y, Kita T, Iguchi R, Hayashi K, Niwa S. De novo mutations in KIF1A-associated neuronal disorder (KAND) dominant-negatively inhibit motor activity and axonal transport of synaptic vesicle precursors. Proc Natl Acad Sci U S A. 2022; 119(32):e2113795119. Epub 2022/08/03. https://doi.org/10. 1073/pnas.2113795119 PMID: 35917346; PubMed Central PMCID: PMC9371658.
- Boyden ED, Campos-Xavier AB, Kalamajski S, Cameron TL, Suarez P, Tanackovic G, et al. Recurrent dominant mutations affecting two adjacent residues in the motor domain of the monomeric kinesin KIF22 result in skeletal dysplasia and joint laxity. Am J Hum Genet. 2011; 89(6):767–72. Epub 2011/ 12/14. https://doi.org/10.1016/j.ajhg.2011.10.016 PMID: 22152678; PubMed Central PMCID: PMC3234368.
- Cheon CK, Lim SH, Kim YM, Kim D, Lee NY, Yoon TS, et al. Autosomal dominant transmission of complicated hereditary spastic paraplegia due to a dominant negative mutation of KIF1A, SPG30 gene. Sci Rep. 2017; 7(1):12527. Epub 2017/10/04. https://doi.org/10.1038/s41598-017-12999-9 PMID: 28970574; PubMed Central PMCID: PMC5624960.
- Duis J, Dean S, Applegate C, Harper A, Xiao R, He W, et al. KIF5A mutations cause an infantile onset phenotype including severe myoclonus with evidence of mitochondrial dysfunction. Ann Neurol. 2016; 80(4):633–7. Epub 2016/07/28. <u>https://doi.org/10.1002/ana.24744</u> PMID: <u>27463701</u>; PubMed Central PMCID: PMC5042851.
- Kalantari S, Carlston C, Alsaleh N, Abdel-Salam GMH, Alkuraya F, Kato M, et al. Expanding the KIF4A-associated phenotype. Am J Med Genet A. 2021; 185(12):3728–39. Epub 2021/08/05. https:// doi.org/10.1002/ajmg.a.62443 PMID: 34346154; PubMed Central PMCID: PMC9291479.
- Min BJ, Kim N, Chung T, Kim OH, Nishimura G, Chung CY, et al. Whole-exome sequencing identifies mutations of KIF22 in spondyloepimetaphyseal dysplasia with joint laxity, leptodactylic type. Am J Hum Genet. 2011; 89(6):760–6. Epub 2011/12/14. https://doi.org/10.1016/j.ajhg.2011.10.015 PMID: 22152677; PubMed Central PMCID: PMC3234366.
- Reilly ML, Ain NU, Muurinen M, Tata A, Huber C, Simon M, et al. Biallelic KIF24 Variants Are Responsible for a Spectrum of Skeletal Disorders Ranging From Lethal Skeletal Ciliopathy to Severe Acromesomelic Dysplasia. J Bone Miner Res. 2022; 37(9):1642–52. Epub 2022/06/25. https://doi.org/10. 1002/jbmr.4639 PMID: 35748595; PubMed Central PMCID: PMC9545074.
- Itai T, Wang Z, Nishimura G, Ohashi H, Guo L, Wakano Y, et al. De novo heterozygous variants in KIF5B cause kyphomelic dysplasia. Clin Genet. 2022; 102(1):3–11. Epub 2022/03/29. <u>https://doi.org/ 10.1111/cge.14133 PMID: 35342932.</u>
- Flex E, Albadri S, Radio FC, Cecchetti S, Lauri A, Priolo M, et al. Dominantly acting KIF5B variants with pleiotropic cellular consequences cause variable clinical phenotypes. Hum Mol Genet. 2023; 32 (3):473–88. Epub 2022/08/27. https://doi.org/10.1093/hmg/ddac213 PMID: <u>36018820</u>; PubMed Central PMCID: PMC9851748.
- Lek M, Karczewski KJ, Minikel EV, Samocha KE, Banks E, Fennell T, et al. Analysis of protein-coding genetic variation in 60,706 humans. Nature. 2016; 536(7616):285–91. Epub 2016/08/19. <u>https://doi.org/10.1038/nature19057</u> PMID: 27535533; PubMed Central PMCID: PMC5018207.
- Au V, Li-Leger E, Raymant G, Flibotte S, Chen G, Martin K, et al. CRISPR/Cas9 Methodology for the Generation of Knockout Deletions in Caenorhabditis elegans. G3 (Bethesda). 2019; 9(1):135–44. Epub 2018/11/14. <u>https://doi.org/10.1534/g3.118.200778</u> PMID: <u>30420468</u>; PubMed Central PMCID: PMC6325907.
- Muller HJ. Further studies on the nature and causes of gene mutations. Proceedings of the Sixth International Congress of Genetics. 1932; 1:213–55.
- Herskowitz I. Functional inactivation of genes by dominant negative mutations. Nature. 1987; 329 (6136):219–22. Epub 1987/09/17. https://doi.org/10.1038/329219a0 PMID: 2442619.

- Nonet M. Additional Landing Sites for Recombination-Mediated Cassette Exchange in C. elegans. MicroPubl Biol. 2021;2021. Epub 2021/12/23. https://doi.org/10.17912/micropub.biology.000503 PMID: 34934910; PubMed Central PMCID: PMC8678631.
- Nonet ML. Efficient Transgenesis in Caenorhabditis elegans Using Flp Recombinase-Mediated Cassette Exchange. Genetics. 2020; 215(4):903–21. Epub 2020/06/10. https://doi.org/10.1534/genetics. 120.303388 PMID: 32513816; PubMed Central PMCID: PMC7404237.
- Mondal S, Ahlawat S, Koushika SP. Simple microfluidic devices for in vivo imaging of C. elegans, Drosophila and zebrafish. J Vis Exp. 2012;(67). Epub 2012/10/12. https://doi.org/10.3791/3780 PMID: 23051668; PubMed Central PMCID: PMC3490237.
- Neumann S, Chassefeyre R, Campbell GE, Encalada SE. KymoAnalyzer: a software tool for the quantitative analysis of intracellular transport in neurons. Traffic. 2017; 18(1):71–88. Epub 2016/10/23. https://doi.org/10.1111/tra.12456 PMID: 27770501; PubMed Central PMCID: PMC5473519.
- Novas R, Cardenas-Rodriguez M, Lepanto P, Fabregat M, Rodao M, Fariello MI, et al. Kinesin 1 regulates cilia length through an interaction with the Bardet-Biedl syndrome related protein CCDC28B. Sci Rep. 2018; 8(1):3019. Epub 2018/02/16. https://doi.org/10.1038/s41598-018-21329-6 PMID: 29445114; PubMed Central PMCID: PMC5813027.
- Dutta P, Ray K. Ciliary membrane, localised lipid modification and cilia function. J Cell Physiol. 2022; 237(6):2613–31. Epub 2022/06/07. https://doi.org/10.1002/jcp.30787 PMID: 35661356.
- Li S, Zhang H, Sun Y. Primary cilia in hard tissue development and diseases. Front Med. 2021; 15 (5):657–78. Epub 2021/09/14. https://doi.org/10.1007/s11684-021-0829-6 PMID: 34515939.
- Masson J, El Ghouzzi V. Golgi Dysfunctions in Ciliopathies. Cells. 2022; 11(18). Epub 2022/09/24. https://doi.org/10.3390/cells11182773 PMID: 36139347; PubMed Central PMCID: PMC9496873.
- Murdock DR, Dai H, Burrage LC, Rosenfeld JA, Ketkar S, Muller MF, et al. Transcriptome-directed analysis for Mendelian disease diagnosis overcomes limitations of conventional genomic testing. J Clin Invest. 2021; 131(1). Epub 2020/10/02. https://doi.org/10.1172/JCI141500 PMID: 33001864; PubMed Central PMCID: PMC7773386.
- Chen J, Ou Y, Luo R, Wang J, Wang D, Guan J, et al. SAR1B senses leucine levels to regulate mTORC1 signalling. Nature. 2021; 596(7871):281–4. Epub 2021/07/23. https://doi.org/10.1038/ s41586-021-03768-w PMID: 34290409.
- Lynch CJ. Role of leucine in the regulation of mTOR by amino acids: revelations from structure-activity studies. J Nutr. 2001; 131(3):861S–5S. Epub 2001/03/10. https://doi.org/10.1093/jn/131.3.861S PMID: 11238775.
- Son SM, Park SJ, Lee H, Siddiqi F, Lee JE, Menzies FM, et al. Leucine Signals to mTORC1 via Its Metabolite Acetyl-Coenzyme A. Cell Metab. 2019; 29(1):192–201 e7. Epub 2018/09/11. https://doi. org/10.1016/j.cmet.2018.08.013 PMID: 30197302; PubMed Central PMCID: PMC6331339.
- Gupta V, Palmer KJ, Spence P, Hudson A, Stephens DJ. Kinesin-1 (uKHC/KIF5B) is required for bidirectional motility of ER exit sites and efficient ER-to-Golgi transport. Traffic. 2008; 9(11):1850–66. Epub 2008/09/27. https://doi.org/10.1111/j.1600-0854.2008.00811.x PMID: 18817524.
- Cao L, Cantos-Fernandes S, Gigant B. The structural switch of nucleotide-free kinesin. Sci Rep. 2017; 7:42558. Epub 2017/02/15. <u>https://doi.org/10.1038/srep42558</u> PMID: <u>28195215</u>; PubMed Central PMCID: PMC5307337.
- Cao L, Wang W, Jiang Q, Wang C, Knossow M, Gigant B. The structure of apo-kinesin bound to tubulin links the nucleotide cycle to movement. Nat Commun. 2014; 5:5364. Epub 2014/11/15. <u>https://doi.org/10.1038/ncomms6364</u> PMID: 25395082.
- Nakata T, Hirokawa N. Point mutation of adenosine triphosphate-binding motif generated rigor kinesin that selectively blocks anterograde lysosome membrane transport. J Cell Biol. 1995; 131(4):1039–53. Epub 1995/11/01. <u>https://doi.org/10.1083/jcb.131.4.1039</u> PMID: <u>7490281</u>; PubMed Central PMCID: PMC2200001.
- Shang Z, Zhou K, Xu C, Csencsits R, Cochran JC, Sindelar CV. High-resolution structures of kinesin on microtubules provide a basis for nucleotide-gated force-generation. Elife. 2014; 3:e04686. Epub 2014/11/22. https://doi.org/10.7554/eLife.04686 PMID: 25415053; PubMed Central PMCID: PMC4383081.
- Boyle L, Rao L, Kaur S, Fan X, Mebane C, Hamm L, et al. Genotype and defects in microtubule-based motility correlate with clinical severity in KIF1A-associated neurological disorder. HGG Adv. 2021; 2 (2). Epub 2021/04/22. https://doi.org/10.1016/j.xhgg.2021.100026 PMID: <u>33880452</u>; PubMed Central PMCID: PMC8054982.
- Pandruvada SN, Beauregard J, Benjannet S, Pata M, Lazure C, Seidah NG, et al. Role of Ostm1 Cytosolic Complex with Kinesin 5B in Intracellular Dispersion and Trafficking. Mol Cell Biol. 2016; 36 (3):507–21. Epub 2015/11/26. https://doi.org/10.1128/MCB.00656-15 PMID: 26598607; PubMed Central PMCID: PMC4719428.

- Claeys L, Storoni S, Eekhoff M, Elting M, Wisse L, Pals G, et al. Collagen transport and related pathways in Osteogenesis Imperfecta. Hum Genet. 2021; 140(8):1121–41. Epub 2021/06/26. https://doi.org/10.1007/s00439-021-02302-2 PMID: 34169326; PubMed Central PMCID: PMC8263409 conflict of interest.
- Lekszas C, Foresti O, Raote I, Liedtke D, Konig EM, Nanda I, et al. Biallelic TANGO1 mutations cause a novel syndromal disease due to hampered cellular collagen secretion. Elife. 2020; 9. Epub 2020/02/ 27. https://doi.org/10.7554/eLife.51319 PMID: 32101163; PubMed Central PMCID: PMC7062462.
- Marom R, Burrage LC, Venditti R, Clement A, Blanco-Sanchez B, Jain M, et al. COPB2 loss of function causes a coatopathy with osteoporosis and developmental delay. Am J Hum Genet. 2021; 108 (9):1710–24. Epub 2021/08/28. https://doi.org/10.1016/j.ajhg.2021.08.002 PMID: 34450031; PubMed Central PMCID: PMC8456174.
- He S, Xue W, Duan Z, Sun Q, Li X, Gan H, et al. Multimodal nonlinear optical microscopy reveals critical role of kinesin-1 in cartilage development. Biomed Opt Express. 2017; 8(3):1771–82. Epub 2017/ 07/01. https://doi.org/10.1364/BOE.8.001771 PMID: 28663865; PubMed Central PMCID: PMC5480580.
- Bergen DJM, Stevenson NL, Skinner REH, Stephens DJ, Hammond CL. The Golgi matrix protein giantin is required for normal cilia function in zebrafish. Biol Open. 2017; 6(8):1180–9. Epub 2017/05/ 27. https://doi.org/10.1242/bio.025502 PMID: 28546340; PubMed Central PMCID: PMC5576078.
- 53. Symoens S, Barnes AM, Gistelinck C, Malfait F, Guillemyn B, Steyaert W, et al. Genetic Defects in TAPT1 Disrupt Ciliogenesis and Cause a Complex Lethal Osteochondrodysplasia. Am J Hum Genet. 2015; 97(4):521–34. Epub 2015/09/15. https://doi.org/10.1016/j.ajhg.2015.08.009 PMID: 26365339; PubMed Central PMCID: PMC4596895.
- 54. Farhan H, Rabouille C. Signalling to and from the secretory pathway. J Cell Sci. 2011; 124(Pt 2):171– 80. Epub 2010/12/29. https://doi.org/10.1242/jcs.076455 PMID: 21187344.
- 55. Chia J, Goh G, Racine V, Ng S, Kumar P, Bard F. RNAi screening reveals a large signaling network controlling the Golgi apparatus in human cells. Mol Syst Biol. 2012; 8:629. Epub 2012/12/06. https://doi.org/10.1038/msb.2012.59 PMID: 23212246; PubMed Central PMCID: PMC3542528.
- 56. Cardoso CM, Groth-Pedersen L, Hoyer-Hansen M, Kirkegaard T, Corcelle E, Andersen JS, et al. Depletion of kinesin 5B affects lysosomal distribution and stability and induces peri-nuclear accumulation of autophagosomes in cancer cells. PLoS One. 2009; 4(2):e4424. Epub 2009/02/27. https://doi. org/10.1371/journal.pone.0004424 PMID: 19242560; PubMed Central PMCID: PMC2647799.
- Du W, Su QP, Chen Y, Zhu Y, Jiang D, Rong Y, et al. Kinesin 1 Drives Autolysosome Tubulation. Dev Cell. 2016; 37(4):326–36. Epub 2016/05/25. <u>https://doi.org/10.1016/j.devcel.2016.04.014</u> PMID: 27219061.
- Jia R, Bonifacino JS. Lysosome Positioning Influences mTORC2 and AKT Signaling. Mol Cell. 2019; 75(1):26–38 e3. Epub 2019/05/28. https://doi.org/10.1016/j.molcel.2019.05.009 PMID: 31130364; PubMed Central PMCID: PMC7446307.
- Jia R, Guardia CM, Pu J, Chen Y, Bonifacino JS. BORC coordinates encounter and fusion of lysosomes with autophagosomes. Autophagy. 2017; 13(10):1648–63. Epub 2017/08/22. https://doi.org/ 10.1080/15548627.2017.1343768 PMID: 28825857; PubMed Central PMCID: PMC5640200.
- Gosavi P, Houghton FJ, McMillan PJ, Hanssen E, Gleeson PA. The Golgi ribbon in mammalian cells negatively regulates autophagy by modulating mTOR activity. J Cell Sci. 2018; 131(3). Epub 2018/01/ 24. https://doi.org/10.1242/jcs.211987 PMID: 29361552.
- Lai Y, Jiang Y. Reciprocal Regulation between Primary Cilia and mTORC1. Genes (Basel). 2020; 11 (6). Epub 2020/07/02. <u>https://doi.org/10.3390/genes11060711</u> PMID: <u>32604881</u>; PubMed Central PMCID: PMC7349257.
- Payne EM, Virgilio M, Narla A, Sun H, Levine M, Paw BH, et al. L-Leucine improves the anemia and developmental defects associated with Diamond-Blackfan anemia and del(5q) MDS by activating the mTOR pathway. Blood. 2012; 120(11):2214–24. Epub 2012/06/27. https://doi.org/10.1182/blood-2011-10-382986 PMID: 22734070; PubMed Central PMCID: PMC3447780.
- Vlachos A, Atsidaftos E, Lababidi ML, Muir E, Rogers ZR, Alhushki W, et al. L-leucine improves anemia and growth in patients with transfusion-dependent Diamond-Blackfan anemia: Results from a multicenter pilot phase I/II study from the Diamond-Blackfan Anemia Registry. Pediatr Blood Cancer. 2020; 67(12):e28748. Epub 2020/10/08. https://doi.org/10.1002/pbc.28748 PMID: 33025707; PubMed Central PMCID: PMC8273758.
- **64.** Xu B, Gogol M, Gaudenz K, Gerton JL. Improved transcription and translation with L-leucine stimulation of mTORC1 in Roberts syndrome. BMC Genomics. 2016; 17:25. Epub 2016/01/06. https://doi. org/10.1186/s12864-015-2354-y PMID: 26729373; PubMed Central PMCID: PMC4700579.

- Chen J, Long F. mTOR signaling in skeletal development and disease. Bone Res. 2018; 6:1. Epub 2018/02/10. https://doi.org/10.1038/s41413-017-0004-5 PMID: 29423330; PubMed Central PMCID: PMC5802487.
- 66. Hanagata N, Takemura T, Kamimura K, Koda T. Effect of immunosuppressants on a mouse model of osteogenesis imperfecta type V harboring a heterozygous lfitm5 c.-14C > T mutation. Sci Rep. 2020; 10(1):21197. Epub 2020/12/05. https://doi.org/10.1038/s41598-020-78403-1 PMID: <u>33273604</u>; PubMed Central PMCID: PMC7713238.
- Joeng KS, Lee YC, Lim J, Chen Y, Jiang MM, Munivez E, et al. Osteocyte-specific WNT1 regulates osteoblast function during bone homeostasis. J Clin Invest. 2017; 127(7):2678–88. Epub 2017/06/20. https://doi.org/10.1172/JCI92617 PMID: 28628032; PubMed Central PMCID: PMC5490765 exists.
- Bainbridge MN, Wang M, Wu Y, Newsham I, Muzny DM, Jefferies JL, et al. Targeted enrichment beyond the consensus coding DNA sequence exome reveals exons with higher variant densities. Genome Biol. 2011; 12(7):R68. <u>https://doi.org/10.1186/gb-2011-12-7-r68</u> PMID: <u>21787409</u>; PubMed Central PMCID: PMC3218830.
- Challis D, Yu J, Evani US, Jackson AR, Paithankar S, Coarfa C, et al. An integrative variant analysis suite for whole exome next-generation sequencing data. BMC Bioinformatics. 2012; 13:8. <u>https://doi.org/10.1186/1471-2105-13-8</u> PMID: 22239737; PubMed Central PMCID: PMC3292476.
- Reid JG, Carroll A, Veeraraghavan N, Dahdouli M, Sundquist A, English A, et al. Launching genomics into the cloud: deployment of Mercury, a next generation sequence analysis pipeline. BMC Bioinformatics. 2014; 15:30. https://doi.org/10.1186/1471-2105-15-30 PMID: 24475911; PubMed Central PMCID: PMC3922167.
- Consortium GT, Laboratory DA, Coordinating Center -Analysis Working G, Statistical Methods groups-Analysis Working G, Enhancing Gg, Fund NIHC, et al. Genetic effects on gene expression across human tissues. Nature. 2017; 550(7675):204–13. <u>https://doi.org/10.1038/nature24277</u> PMID: 29022597; PubMed Central PMCID: PMC5776756.
- 72. Dobin A, Davis CA, Schlesinger F, Drenkow J, Zaleski C, Jha S, et al. STAR: ultrafast universal RNAseq aligner. Bioinformatics. 2013; 29(1):15–21. https://doi.org/10.1093/bioinformatics/bts635 PMID: 23104886; PubMed Central PMCID: PMC3530905.
- Li B, Dewey CN. RSEM: accurate transcript quantification from RNA-Seq data with or without a reference genome. BMC Bioinformatics. 2011; 12:323. <u>https://doi.org/10.1186/1471-2105-12-323</u> PMID: 21816040; PubMed Central PMCID: PMC3163565.
- 74. Robinson JT, Thorvaldsdottir H, Winckler W, Guttman M, Lander ES, Getz G, et al. Integrative genomics viewer. Nat Biotechnol. 2011; 29(1):24–6. <u>https://doi.org/10.1038/nbt.1754</u> PMID: <u>21221095</u>; PubMed Central PMCID: PMC3346182.
- 75. Schymkowitz J, Borg J, Stricher F, Nys R, Rousseau F, Serrano L. The FoldX web server: an online force field. Nucleic Acids Res. 2005; 33(Web Server issue):W382–8. Epub 2005/06/28. <u>https://doi.org/10.1093/nar/gki387 PMID: 15980494</u>; PubMed Central PMCID: PMC1160148.
- 76. Hanna SL, Sherman NE, Kinter MT, Goldberg JB. Comparison of proteins expressed by Pseudomonas aeruginosa strains representing initial and chronic isolates from a cystic fibrosis patient: an analysis by 2-D gel electrophoresis and capillary column liquid chromatography-tandem mass spectrometry. Microbiology (Reading). 2000; 146 (Pt 10):2495–508. Epub 2000/10/06. https://doi.org/ 10.1099/00221287-146-10-2495 PMID: 11021925.
- Fielder SM, Rosenfeld JA, Burrage LC, Emrick L, Lalani S, Attali R, et al. Functional analysis of a novel de novo variant in PPP5C associated with microcephaly, seizures, and developmental delay. Mol Genet Metab. 2022; 136(1):65–73. Epub 2022/04/02. <u>https://doi.org/10.1016/j.ymgme.2022.03.</u> 007 PMID: 35361529.
- Dokshin GA, Ghanta KS, Piscopo KM, Mello CC. Robust Genome Editing with Short Single-Stranded and Long, Partially Single-Stranded DNA Donors in Caenorhabditis elegans. Genetics. 2018; 210 (3):781–7. Epub 2018/09/15. https://doi.org/10.1534/genetics.118.301532 PMID: 30213854; PubMed Central PMCID: PMC6218216.
- Paix A, Folkmann A, Rasoloson D, Seydoux G. High Efficiency, Homology-Directed Genome Editing in Caenorhabditis elegans Using CRISPR-Cas9 Ribonucleoprotein Complexes. Genetics. 2015; 201 (1):47–54. Epub 2015/07/19. https://doi.org/10.1534/genetics.115.179382 PMID: 26187122; PubMed Central PMCID: PMC4566275.
- Dickinson DJ, Ward JD, Reiner DJ, Goldstein B. Engineering the Caenorhabditis elegans genome using Cas9-triggered homologous recombination. Nat Methods. 2013; 10(10):1028–34. Epub 2013/ 09/03. https://doi.org/10.1038/nmeth.2641 PMID: 23995389; PubMed Central PMCID: PMC3905680.
- 81. Huang H, Pan J, Spielberg DR, Hanchard NA, Scott DA, Burrage LC, et al. A dominant negative variant of RAB5B disrupts maturation of surfactant protein B and surfactant protein C. Proc Natl Acad Sci

U S A. 2022; 119(6). Epub 2022/02/06. https://doi.org/10.1073/pnas.2105228119 PMID: 35121658; PubMed Central PMCID: PMC8832968.

- Patro R, Duggal G, Love MI, Irizarry RA, Kingsford C. Salmon provides fast and bias-aware quantification of transcript expression. Nat Methods. 2017; 14(4):417–9. Epub 2017/03/07. https://doi.org/10. 1038/nmeth.4197 PMID: 28263959; PubMed Central PMCID: PMC5600148.
- Charng WL, Karaca E, Coban Akdemir Z, Gambin T, Atik MM, Gu S, et al. Exome sequencing in mostly consanguineous Arab families with neurologic disease provides a high potential molecular diagnosis rate. BMC Med Genomics. 2016; 9(1):42. Epub 2016/07/21. https://doi.org/10.1186/ s12920-016-0208-3 PMID: 27435318; PubMed Central PMCID: PMC4950750.